Author's response to reviews

Title: Postnatal depressive symptoms in Vietnam: Investigating the influence of son preference, traditional confinement practices and other socio-cultural factors

Authors:

Linda J Murray (linda.murray@utas.edu.au)
Michael P Dunne (m.dunne@utas.edu.au)
Thang Vo Van (yovanthang147@gmail.com)
Nguyen Thi An Phuong (thuynguyen201@gmail.com)
Nigar Khawaja (n.khawaja@qut.edu.au)
Thanh Cao Ngoc (thuynguyen201@gmail.com)

Version: 2 Date: 14 November 2014

Author's response to reviews:

Dear Sir Madam,

As requested in your email I have made the following changes to my manuscript:

1. Line numbers have been added to the manuscript
2. Page breaks have been added

Regards
Linda Murray